Author's response to reviews

Title: Antibacterial activity of some selected medicinal plants of Pakistan

Authors:

Yamin Bibi (yamin.phd@gmail.com)
Sobia Nisa (sobianisa_qau@yahoo.com)
Fayyaz M Chaudhary (fayyazchaudhary@yahoo.com)
Zia Muhammad (ziachaudhary@gmail.com)

Version: 8 Date: 30 June 2011

Author's response to reviews: see over
To,

Chief Editor,

BMC Complementary and Alternative medicine

Respected Dear,

I am submitting with this letter a proof reading version of our manuscript “Antibacterial activity of some selected medicinal plants of Pakistan” (MS: 3986070515205425) by Yamin Bibi, Sobia Nisa, Fayyaz M Chaudhary and Zia Muhammad, for publication in BMC Complementary and Alternative Medicine.

We hope our revised manuscript meets the requirements for publication in the BMC Complementary and Alternative Medicine. If you require any further information, please do not hesitate to contact us directly.

With sincere regards,

Dr. Muhammad Zia (Corresponding author)

Lecturer, Department of Biotechnology,

Quaid-i-Azam University, Islamabad Pakistan, 45320

Email: ziachaudhary@gmail.com